$See \ discussions, stats, and author \ profiles \ for \ this \ publication \ at: \ https://www.researchgate.net/publication/349249863$

Little Ice Age Revealed in Tree-Ring-Based Precipitation Record From the Northwest Himalaya, India

Article in Geophysical Research Letters · February 2021

DOI: 10.1029/2020GL091298

Project

Project

IONS		reads 280	
thors, including:			
Vikram Singh		K. G. Misra	
Birbal Sahni I	nstitute of Palaeobotany	Birbal Sahni	Institute of Palaeosciences
10 PUBLICATION	IS 74 CITATIONS	28 PUBLICATIO	DNS 236 CITATIONS
SEE PROFILE		SEE PROFILI	E
Arun Deo Sin	gh	Ram Yadav	
Banaras Hind	u University	Birbal Sahni	Institute of Palaeobotany
113 PUBLICATIO	NS 1,361 CITATIONS	57 PUBLICATIO	DNS 1,879 CITATIONS
SEE PROFILE		SEE PROFILI	E

Some of the authors of this publication are also working on these related projects:

Analysis of plants and human relationsip since late pleistocene from dzukou valley, Nagaland northeast india View project

Palaeoclimatology of western Himalaya View project



Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL091298

Key Points:

- Three distinct centennial scale phases have been identified in 635 years long precipitation record developed from the northwest Himalaya
- Stable precipitation during 1650s-1850s CE reveals clear evidence of Little Ice Age (LIA) for the north-western Himalaya
- High-magnitude droughts or low precipitation limit the LIA influence before 1650s CE

Supporting Information:

Supporting Information S1

Correspondence to:

V. Singh and K. G. Misra, vnegi850@gmail.com; kgmisrabsip@gmail.com

Citation:

Singh, V., Misra, K. G., Singh, A. D., Yadav, R. R., & Yadava, A. K. (2021). Little Ice Age revealed in tree-ringbased precipitation record from the northwest Himalaya, India. *Geophysical Research Letters*, *48*, e2020GL091298. https://doi.org/10.1029/2020GL091298

Received 17 NOV 2020 Accepted 3 FEB 2021

Author Contributions:

Conceptualization: Vikram Singh, Krishna G. Misra, Ram R. Yadav Formal analysis: Vikram Singh Investigation: Vikram Singh, Krishna G. Misra, Arun D. Singh, Ram R. Yadav, Akhilesh K. Yadava Supervision: Krishna G. Misra, Arun D. Singh, Ram R. Yadav Writing – original draft: Vikram Singh Writing – review & editing: Vikram Singh, Krishna G. Misra, Arun D. Singh, Ram R. Yadav, Akhilesh K. Yadava

© 2021. American Geophysical Union. All Rights Reserved.

Little Ice Age Revealed in Tree-Ring-Based Precipitation Record From the Northwest Himalaya, India

Vikram Singh^{1,2}, Krishna G. Misra¹, Arun D. Singh^{2,3}, Ram R. Yadav⁴, and Akhilesh K. Yadava¹

¹Birbal Sahni Institute of Palaeosciences, Lucknow, India, ²Department of Geology, Centre of Advanced Study, Institute of Science, Banaras Hindu University, Varanasi, India, ³DST-Mahamana Centre of Excellence in Climate Change Research, Institute of Environment and Sustainable Development, Banaras Hindu University, Varanasi, India, ⁴Wadia Institute of Himalayan Geology, Dehradun, India

Abstract The spatial and temporal span of hydrological impact of the "Little Ice Age" (LIA) in the north-western Himalaya is not well constrained due to data limitation. We evaluated a network of treering chronologies from moisture-stressed ecological settings in Jammu and Kashmir to identify the impact of LIA over the western Himalaya. Our study reflects three centennial scale phases; the middle phase (1650s–1850s CE) with stable precipitation and in accordance with other hydrological records clearly demarcates the LIA impact over the Himalaya. However, high-magnitude droughts recorded in early (1383–1650s CE) and late phases (1850s–2017) underpin that the north-western Himalaya did not witness any LIA influence before 1650s.

Plain Language Summary Here, we present a 635-year long annual (prior year October to current year September) precipitation reconstruction from Kishtwar, Jammu and Kashmir. A network of 16 tree-ring chronologies of Himalayan cedar and neoza pine was used in the study. The reconstruction captured three distinct phases in precipitation regime from 1383–1650s, 1650s–1850s, and 1850s–2017 CE and among these, the middle phase (1650s–1850s) showed relatively prolonged pluvial phase with consistently stable precipitation advocating for the active LIA over the western Himalaya. The other two phases reflect high-amplitude droughts with intervening wet periods contrasting the existence of the LIA-induced wetting. To validate the reconstruction skill and the LIA effect over the western Himalaya and Central Asia, the present record was compared with the other available precipitation, river flow and hydrological proxy records which support the LIA influenced precipitation during 1650s–1850s CE.

1. Introduction

The Little Ice Age (LIA) is marked by the glacial expansion and widespread cooling in the Northern Hemisphere (Mann et al., 2009), while the timing of its advancement in different continents was spatially varied (Rowan, 2017). The continental scale temperature reconstructions (PAGES 2k Consortium, 2013) also recorded variable pattern in temperature in the last two millennia with distinct asynchronicity in multi-decadal warm and cold events, which could define Medieval Climate Anomaly and LIA worldwide. The LIA period is generally defined from fifteenth to nineteenth century with about 0.6°C temperature drop over the Northern Hemisphere (Bradley & Jones, 1993; Jones et al., 1998; Mann, 2002; Mann et al., 1998, 1999). However, Arctic Canada and Iceland ice-cap growth reflects the cold summer and ice growth started abruptly during 1275-1300 CE that coincided with the active volcanic period of the past millennium (Miller et al., 2012), whereas Europe was influenced largely during sixteenth to mid-nineteenth century (Mann, 2002). The LIA has major significance in the determination of global climate of the past millennium, but still controversies regarding its initiation, duration, and spatial advancements continue to exist across the globe. The one belief is that the LIA-induced glacial expansion occurred in north of 20°N areas which largely depends on the winter precipitation, while summer temperature was not the sole factor for LIA climate (Matthews & Briffa, 2005). However, another belief is that the glaciers expanded throughout the globe (Grove, 2001; Porter, 1981) and temperature fall had significant contribution in the LIA (Hughes & Diaz, 1994; Grove, 2001; Mann, 2002; Nesje & Dahl, 2003). Such anomalous results are likely due to a lack of precise environmental records from across the world restricting our understanding of LIA and its impact.

The Himalaya has the largest area of snow/ice outside the poles, but instrumental as well as proxy records are lacking to explain exact temporal and spatial influence of LIA. Although in recent decades, several specific studies on mass balance targeted to identify the past glacial history of the Himalayan glaciers have been utilized (Hedrick et al., 2011; Lehmkuhl et al., 1998; Kulkarni & Alex, 2003; Kulkarni & Rathore, 2003; Kulkarni et al., 2010; Murari et al., 2014; Owen et al., 2000, 2001, 2002, 2005; Rowan, 2017; Solomina et al., 2015). These studies also reflect variability in the timing and extent of glacial advance in the Himalaya during the LIA. A recent study (Managave et al., 2020) based on scPDSI denoting contrasting hydro-climatic trends in the western Himalaya and Karakoram region, revealed that the peak glacial activity occurred around 1300–1600 CE in the Lahaul region of the western Himalaya. However, a tree-ring-based glacial mass balance reconstruction from the Himalaya revealed significant mass loss during the LIA in the Himalaya (Shekhar et al., 2017). Such heterogeneity in proxy records highlights the role of regional characteristics of climatic phenomena in the orography-dominated Himalaya during the LIA. Although the highly dissected and orography-dominated Himalaya has large-scale spatial variability, inconsistent and patchy climatic records are inadequate to capture past climatic upheavals (Cook et al., 2010; Yadava et al., 2016).

The present study was carried out to investigate precipitation variability during the LIA in the western Himalaya. For this, tree-ring data from moisture-stressed sites in Jammu and Kashmir were used. The state of Jammu and Kashmir has large glaciated area where westerlies first strike the Indian subcontinent with moisture transported from the Mediterranean and Atlantic Ocean. Limited tree-ring-based hydrological studies have been carried out so far from Jammu and Kashmir (Ballesteros-Canovas et al., 2020; Borgaonkar et al., 1994; Hughes, 1992, 2001; Shah et al., 2018; Singh et al., 2017; Yadav et al., 2017). Here, we report a 635-year long precipitation record, the longest so far from this region, and highlight its linkage with the LIA over the western Himalaya.

2. Data and Methods

2.1. Tree-Ring Data

Tree-ring materials comprising 525 increment cores from 339 trees (356 increment cores from 212 neoza pine; Pinus gerardiana Wall. ex Lamb.) and 169 cores from 127 Himalayan cedar (Cedrus deodara [Roxb.] G. Don) were collected during three consecutive field excursions (2015, 2017 and 2018) from Kishtwar, Jammu and Kashmir. Sixteen (nine of neoza pine and seven of Himalayan cedar) homogeneous and moisture-stressed sites (Figure 1) were selected for the present study. Generally, two cores were extracted from each tree at breast height (~1.4 m) in the direction perpendicular to the slope and processed in the laboratory following the standard dendrochronological methods (Stokes & Smiley, 1968). The ring-width sequences were measured using the LINTAB (Rinntech) measuring machine, with a resolution of 0.01 mm (Rinn, 2003) and ring-width measurements of precisely dated samples were cross-checked in the COFE-CHA (Holmes, 1983) program. Strong correlation and high similarity in both species highlight the common climate forcing regulating the tree growth over the region. The method opted for chronology development and site details, number of samples, chronology span, and Expressed Population Signal are given in supporting information (Table S1). Out of the total 16 chronologies, one Himalayan cedar and four neoza pine chronologies are common in present and earlier studies (Singh et al., 2017; Yadav et al., 2017). High correlation and strong year-to-year coherence among all the chronologies for the common period 1842-2014 (Table S2) indicate the dominant influence of common climate forcing on tree growth over the study sites.

2.2. Climate Data and its Relationship With Tree-Growth

The instrumental records over the high-altitude Himalaya are very short and patchy. The available records show high spatial variability and therefore, single-station climate data are not usually suitable in dendroclimatic modeling (Singh et al., 2017; Yadav, 2011a, 2011b; Yadav et al., 2004, 2011, 2017). To minimize the effect of single-station data errors, seven meteorological station data were used to prepare regional precipitation series (Figure S1; Table S3). The first principal component of seven-station precipitation data (January–December) for the common period 1954–1994 explains 66% common variance and shows strong coherence (Figure S1; Table S4). The monthly precipitation data were normalized with respect to mean and standard deviation values of the common period and merged to prepare a regional series. The regional





Figure 1. (a) General view of the study area with location of the meteorological stations and the hydrological proxy records used, (b) location of tree-ring sampling site over Kishtwar, Jammu and Kashmir.

precipitation series were then rescaled to the actual values relative to the mean and standard deviation of respective months of all the stations and used in dendroclimatic modeling. In the absence of temperature data from Kishtwar, Srinagar data (34°05′N, 74°48′E, 1585 m asl) ranging from 1893 to 2008 have been used to understand the climate and tree-growth relationship.

The correlation among climatic variables and first principal component (PC#1) of ring-width chronologies having eigenvalue greater than 1 that represents 84.28% variance of the chronologies for the common period 1842–2014 CE, showed positive correlation with precipitation from previous year October to current year September and negative correlation for the whole year temperature except current year September (Figure S2). The significant positive correlation in PC#1 and regional precipitation was recorded in December of the prior year and March, May and July of the current year, while for temperature significant negative correlation was found in October of the prior year and in January, March, April, May, June and July of the current year (Figure S2).

2.3. Precipitation Reconstruction

Annual precipitation from prior year October to current year September (pOcS) was reconstructed using a network of 16 chronologies from Kishtwar, Jammu and Kashmir. To optimize the reconstruction length, the nested approach (Cook et al., 2003; Meko, 1997) was used in which the chronologies with decreasing length were removed from the pool of the larger set. Seventeen nested series were developed and the PC#1 of each nest was used in the calibration, verification, and reconstruction. The split-period calibration, verification for the two subperiods 1954–1973 and 1974–1994 using regional annual precipitation series and leave-one-out cross-validation method (Michaelsen, 1987) was performed to test the fidelity of calibration models used in the reconstruction. The statistical parameters in calibration and verification analysis such as reduction of error (RE), coefficient of efficiency (CE), Sign test and Pearson correlation coefficient (Cook et al., 1999; Fritts, 1976) were tested (Table S5). Positive values of RE and CE express the statistical skill in the annual (pOcS) precipitation reconstruction. The whole period calibration model 1954-1994 was used in the final reconstruction and mean series developed by combining the individual reconstructions. However, before averaging the values, the nested reconstruction series were scaled with mean and standard deviation of the most-replicated nest to minimize the artefact associated with the change in variance. The mean reconstructed precipitation series extends back to 1383 CE and revealed a significant correlation and close year-to-year similarity with the actual regional precipitation series (Figure S3; r = 0.64, p < 0.0001, 1954-1994).

3. Discussion and Conclusions

The reconstructed *pOcS* precipitation series (1383–2017 CE) reflects high year-to-year and interdecadal to centennial-scale variability (Figure 2). The most notable feature in the reconstructed 635 years data is three distinct centennial phases, early (1383–1650s) and late phase (1850s–2017) reflect relatively unstable and more turbulence but the middle phase (1650s–1850s) revealed stable conditions where the precipitation centered close to the long term mean. The middle phase reflects LIA-induced stable precipitation and its significant influence over the western Himalaya, while the high-magnitude drought or low precipitation periods in the early and late phases do not advocate for the LIA influence over the northwest Himalaya before 1650 CE. The low precipitation during the fifteenth and early-sixteenth centuries noted in the present reconstruction showed resemblance with the March–June precipitation record of the western Himalaya (Yadav, 2011a), where extensive dry periods were recorded during 15th and 16th centuries with 1410–1510 being the driest period. The extended low precipitation period also showed strong similarity with δ^{18} O treering derived precipitation reconstruction for south-western Asia (Anderson et al., 2002). Besides, Anderson et al. (2002) also showed pluvial conditions in the mid-seventeenth to mid-nineteenth centuries consistent with the stable precipitation of present reconstruction.

Similarly, the western Himalaya temperature record (Yadav et al., 2011) also reflected 18th and 19th centuries being the coldest in last millennium when glaciers expanded over the region (Figure 2e). Based on the present study, we believe that cooling in the 18th and 19th centuries (Yadav et al., 2011) could be associated with the LIA-linked stable precipitation. The reconstructed pOcS precipitation series showed significant negative correlation with reconstructed temperature (Yadav et al., 2011) for the common period 1383-2008 CE (r = -0.33, p < 0.0001). Similar to the summer temperature, significant negative correlation was also observed with the spring (March–May) temperature (Yadav et al., 2004) of western Himalaya (r = -0.416, p < 0.00001, 1383-2000 CE; Figure S4). The cool periods during early-eighteenth to mid-nineteenth century were consistent with our data. Based on the glacio-archeological records, settled colonization has been identified within the glacial moraine complex of Tharang, Miyar basin, Lahaul Himalaya at more than 3,500 m asl during the late-tenth to early-nineteenth century (Saini et al., 2019). Saini et al. (2019) also marked the three sites of Tharang, Phundang and Patam (presently abandoned) showing settled agriculture coincident with warm temperature and limited snowfall. During 1300-1600 CE, there was no glacial expansion in the region and the period was generally warmer compared to the present day, which allowed permanent settlements within the end-moraine complex. However, the Miyar basin experienced glacial advances in the period of late-eighteenth and early-nineteenth centuries, which reflect close agreement with the present precipitation record and radiocarbon chronology (Saini et al., 2019). Climate proxies and map records revealed





Figure 2. (a) *pOcS* precipitation reconstruction (1383–2017 CE) with 40-year spline superimposed and the shaded region is the prediction error. The rectangle shows the LIA influenced period of stable precipitation during 1650s–1850s. The comparison of (a) present reconstruction with tree-ring-based precipitation reconstructions of (b) Kinnaur, Himachal Pradesh, (c) Lahaul-Spiti, Himachal Pradesh, (d) Uttarkashi, Uttarakhand and (e) summer temperature of Lahaul-Spiti. All records were standardized with respect to mean and standard deviation of the common period, and stable and major low precipitation periods were highlighted by red and yellow vertical bars, respectively. LIA, Little Ice Age; *pOcS*, prior year October to current year September.

that the settlement abandonment occurred during late-eighteenth and early-nineteenth centuries due to reduced temperature and increased snowfall (Saini et al., 2019). The study of Qiao and Yi (2017) also found a significant role of precipitation in controlling the pattern of glaciation during the LIA over the Himalaya. The stable precipitation recorded in our study, in combination with other proxies, revealed that the LIA in the western Himalaya peaked during 1650s–1850s. Furthermore, the November–April, Snow Water Equivalent (SWE) record (Yadav & Bhutiyani, 2013) from the western Himalaya also revealed close similarity with present precipitation (r = 0.621, p < 0.00001, 1460–2008 CE; Figure S5). The low precipitation periods of early and late phases showed strong consistency with the SWE record (Yadav & Bhutiyani, 2013), while the SWE was relatively stable in the middle phase (1650s–1850s CE) similar to the present study.

The regional strength of our reconstruction was further tested by comparing it with other precipitation records of the western Himalaya. Three independent records, boreal spring precipitation (March–May) from Kinnaur, Himachal Pradesh (Yadava et al., 2016), annual precipitation (previous year August to current year July) from cold arid Lahual-Spiti, Himachal Pradesh (Yadav, 2011b) and spring (March–May) precipitation from Uttarkashi, Uttarakhand (Singh et al., 2006) were compared with present precipitation record (Figures 2a–2d). Strong and significant correlation with precipitation records; Yadava et al. (2016) (r = 0.469, p < 0.00001, 1383–2011 CE), Yadav et al. (2011b) (r = 0.525, p < 0.00001, 1383–2008 CE) and Singh et al. (2006) (r = 0.346, p < 0.0001, 1560–1997) endorse the strength of our reconstruction to capture regional scale features. The LIA-induced stable precipitation reflected in the middle part of the reconstruction was present over the whole western Himalaya (Figures 2a–2d).

To understand the past hydrological challenges and conditions, present precipitation reconstruction was also compared with tree-ring-based river flow records of Satluj (Misra et al., 2015; Singh & Yadav, 2013) and Indus (Rao et al., 2018; Figures 3b–3d). The Pearson correlation of present precipitation reconstruction with flow of Satluj (Singh & Yadav, 2013; r = 0.467, p < 0.0001, 1383–2005 CE), (Misra et al., 2015; r = 0.566, p < 0.00001, 1660–2004 CE) and Indus (Rao et al., 2018; r = 0.24, p < 0.0001, 1394–2005 CE), suggest synoptic scale variability in precipitation over the western Himalayan region that directly influenced river discharge. Furthermore, it has been clearly identified that similar to precipitation, river-flow records also showed lower flow in the early and late phases and stable discharge in the middle phase (1650s–1850s CE; Figures 3b–3d). In comparison to the low river flow, relatively stable river discharge in the middle phase also supports the LIA-induced stable precipitation at that time.

Present precipitation was also compared with tree-ring-based hydrological records of Jammu and Kashmir. In our earlier studies (Singh et al., 2017; Yadav et al., 2017), a standardized precipitation index (SPI8-May and SPI2-May) was developed for Kishtwar, Jammu and Kashmir region and among them, the SPI2-May record inferred severe dry climatic conditions for the region from 1439–1660s CE and wetter conditions from 1670s–2014 (Yadav et al., 2017). However, the pluvial condition from 1670s onwards was also interrupted by drier periods during 1850s–1870s and 1970s. These two drought periods, along with three other low-precipitation periods before 1650s also recorded in our data (Figures 3e and 3f) strongly support the finding of the present study. However, the SPI8-May record showed opposite trend with present precipitation during the late-eighteenth century (1760s–1790s CE; Figure 3g). This could be largely due to the use of single site Himalayan cedar chronology with low replication of samples in developing SPI8-May reconstruction (Singh et al., 2017). However, in the present study, we have used 16 chronologies and among them few chronologies, one of Himalayan cedar in SPI8-May and four of neoza pine with one Himalayan cedar in SPI2-May reconstruction are common. A comparison of the present reconstruction with April–May–June precipitation of Lidder Valley, Jammu and Kashmir (Shah et al., 2018) also revealed close similarity (r = 0.493, p < 0.0001, 1723–2010 CE; Figure 3h).

Apart from this, numerous societal challenges/consequences related to glacial advances in Jammu and Kashmir are reported to have occurred during the LIA period. For example, famines of 1814 and 1831–1833 in Kashmir (Anonymous, 1909; Bamzai, 1962; Koul, 1978) are associated with high precipitation and snow-fall. While in the winter of 1841, devastating floods occurred over Indus and Kashmir (Anonymous, 1909). Jammu and Kashmir region experienced pluvial conditions during 1810s–1840s (Singh et al., 2017). A recent study (Ballesteros-Cánovas et al., 2020) revealed a number of recurrent flood events in 16th–18th century over the Kashmir region. Along with the increased flood events, the size of Wular Lake, Kashmir also increased to its maximum extent during the 18th and 19th century (up to about 200 km²) reflecting





Figure 3. (a) Present precipitation reconstruction along with the western Himalayan river discharge; (b), (c) – Satluj flow, (d) – Indus flow in the upper half and lower half representing comparison of (e) present precipitation with hydrological records of Jammu and Kashmir, (f) SPI2-May, (g) SPI8-May and (h) April–May–June precipitation (data digitized from Shah et al., 2018). The stable precipitation along with relatively high-river discharge and low precipitation along with low river discharge is highlighted by red and yellow vertical bars, respectively. SPI, standardized precipitation index.





Figure 4. Comparison of (a) *pOcS* precipitation record with other hydrological proxy records, (b) Chironomid-based salinity from Sugan Lake, a direct measure of aridity (Chen et al., 2009), (c) Synthesized moisture data (Chen et al., 2010), (d) Guliya Ice core accumulation data from the western Kunlun mountains, Northwestern Tibetan Plateau (Thompson et al., 1995, 1997), (e) Bosten lake CaCO₃ content (Chen et al., 2006), (f) Bosten lake Pollen A/C (*Artemisia*/ Chenopodiaceae) moisture index (Chen et al., 2006), (g) Issyk-Kul lake moisture index derived from *Artemisia/Ephedra* Pollen (Giralt et al., 2004) and (h) Chulerasim cave, δ^{18} O Isotope data, Kumaun Lesser Himalaya (Kotlia et al., 2017), of central Asia and the western Himalaya. The data (a)–(h) were normalized relative to the mean and standard deviation of the series and figures (e), (f), and (g) were digitized from published papers. The highlighted section shows the stable precipitation over the region. *pOcS*, prior year October to current year September.

LIA-induced pluvial conditions. However, the Wular Lake constricted in the sixth to seventh, sixteenth to seventeenth and in late-twentieth century (between 55 and 90 km²), highlighting the stressed hydrological conditions before and after the LIA. All these records in combination with the present study suggest that the LIA was active during the mid-seventeenth to mid-nineteenth century over the western Himalaya, when maximum glacial advances and associated climatic events occurred due to consistently stable, good precipitation regime occurred over long period.

We further compared our present precipitation reconstruction with seven independent hydrological records available from the central Asia and Himalaya (Chen et al., 2010; Chen et al., 2009; Chen et al., 2006; Giralt et al., 2004; Kotlia et al., 2017; Thompson et al., 1995, 1997; Figure 4). As most of these proxy records have coarser temporal resolution compared to the tree-rings, a 40-year low-pass filter of tree-ring records was used to highlight low-frequency variations (Figure 4a). Hydrological records from diversified regions (Figures 4b–4h) indicate strong consistency with the stable precipitation during the middle phase and support the LIA influence. The low precipitation in the twentieth century also reflected in other proxy records (Chen et al., 2010; Chen et al., 2009; Chen et al., 2006; Giralt et al., 2004; Kotlia et al., 2017; Thompson et al., 1995, 1997) is consistent with the present precipitation reconstruction. The precipitation record was also compared with MADA (Cook et al., 2010) of a grid close to our study site ($32^\circ30'-35^\circ$ N, $72^\circ30'-75^\circ$ E). The comparison revealed significant correlation (r = 0.313, p < 0.0001, 1383-2005) and showed its utility to understand the regional and seasonal hydroclimatic variability. Major drought periods recorded in the early and late phases of present study also showed consistency with the June–July–August PDSI (Cook et al., 2010) supporting our findings (Figure S6).

The *pOcS* reconstruction revealed five major low precipitation/drought periods in the 1470s–1500s, 1530s–1570s, 1610s–1640s, 1850s–1890s, 1960s–1980s CE (Figure 2) and out of these, three fall during 1383–1650 CE indicating dryness reflecting no impact of LIA during this period. The late-fifteenth century drought (1470s–1500s CE) was consistent with the tree-ring records from north-eastern Tibetan Plateau (Gou et al., 2010; Shao et al., 2005; Zhang et al., 2003), revealing a wide geographical extent of the high-intensity drought. Dry climatic conditions and low precipitation also prevailed in late-fifteenth century droughts during 1610s–1640s CE were concurrent with the colder temperature and severe drought in the Late Ming period (1620s–1640s CE) over China, when agricultural production declined (Zheng et al., 2014). After the end of the LIA, two major drought periods (1850s–1890s and 1960s–1980s) were recorded in the precipitation reconstruction. The 1960s–1980s low precipitation events were coincidental with the change in Bosten Lake, where a 3 m drop in the lake level was observed during 1959–1987 CE, while the rising lake level was found since 1987 CE and exceeded the existing level of 1959 CE in 1999 CE (Chen et al., 2006). We are of the opinion that a more robust and strong network of proxy records would be needed to establish temporal and spatial extent of the LIA in global perspective.

Conflict of Interest

The authors declare no competing interests.

Data Availability Statement

The reconstruction data presented in the study can be found at https://www.ncdc.noaa.gov/paleo/study/32565.

References

Anderson, D. M., Overpeck, J. T., & Gupta, A. K. (2002). Increase in the Asian southwest monsoon during the past four centuries. *Science*, 297, 596–599.

Anonymous (1909). Imperial gazetteer of India: Provisional Series e Kashmir and Jammu (pp. 140). Calcutta: Superintendent of Government Printing.

Ballesteros-Cánovas, J. A., Koul, T., Bashir, A., del Pozo, J. M. B., Allen, S., Guillet, S., et al. (2020). Recent flood hazards in Kashmir put into context with millennium-long historical and tree-ring records. *The Science of the Total Environment*, 722(137875), 1–9.
 Bamzai, P. N. K. (1962). *A history of Kashmir*. Delhi: Metropolitan Book Co.

Acknowledgments

V. Singh, K. G. Misra, and A. K. Yadava sincerely acknowledge to Dr. Vandana Prasad, Director, BSIP, Lucknow for support and allowing publishing this work (BSIP/RDCC No. 59/2020-21). The authors are thankful to the Forest Department, Jammu and Kashmir, India for support during the field and IMD for providing meteorological data. V. Singh and K. G. Misra are thankful to Parth R. Chauhan (IISER, Mohali) for language editing and Jayendra Singh (WIHG, Dehradun), Anoop K. Singh (Lucknow University), Mukund P. Rao (LDEO, Columbia University) for sharing the data. The authors sincerely acknowledge the Editor, Prof. Valerie Trouet and anonymous reviewers for critical comments, which helped to improve the earlier version of the manuscript. This study is supported by the CSIR-New Delhi (09/528(0021)/2018-EMR-I) to V. Singh (CSIR-SRF), K. G. Misra and V. Singh acknowledges to SERB-DST (SB/DGH-76/2013) and R. R. Yadav acknowledges to CSIR-New Delhi for the support under the Emeritus Scientist scheme (No. 21(1010)/15/EMR-II).



Beveridge, A. S. (1921). Babur-Nama (Memoirs of Babur) (Translated from the original Turki text of Zahiru'd-din Muhammad Babur Padshah Ghazi). Delhi: D.K. Fine Art Press Pvt. Ltd.

- Borgaonkar, H. P., Pant, G. B., & Kumar, K. R. (1994). Dendroclimatic reconstruction of summer precipitation at Srinagar, Kashmir, India, since the late-eighteenth century. *The Holocene*, *4*, 299–306.
- Bradley, R. S., & Jones, P. D. (1993). "Little Ice Age" summer temperature variations: Their nature and relevance to recent global warming trends. *The Holocene*, *3*, 367–376.
- Chen, F., Huang, X., Zhang, J., Holmes, J. A., & Chen, J. (2006). Humid Little Ice Age in arid Central Asia documented by Bosten Lake, Xinjiang, China. Science in China – Series D: Earth Sciences, 49, 1280–1290.
- Chen, F. H., Chen, J. H., Holmes, J., Boomer, I., Austin, P., Gates, J. B., et al. (2010). Moisture changes over the last millennium in arid Central Asia: A review, synthesis and comparison with monsoon region. *Quaternary Science Reviews*, 29, 1055–1068.
- Chen, J. H., Chen, F., Zhang, E., Brooks, S. J., Zhou, A. F., & Zhang, J. W. (2009). A 1000-year chironomid-based salinity reconstruction from varved sediments of Sugan Lake, Qaidam Basin, arid Northwest China, and its paleoclimatic significance. *Chinese Science Bulletin*, 54, 3749–3759.
- Cook, E. R., Anchukaitis, K. J., Buckley, B. M., D'Arrigo, R. D., Jacoby, G. C., & Wright, W. E. (2010). Asian monsoon failure and mega drought during the last millennium. *Science*, 328, 486–489.
- Cook, E. R., Krusic, P. J., & Jones, P. D. (2003). Dendroclimatic signals in long tree-ring chronologies from the Himalayas of Nepal. International Journal of Climatology, 23, 707–732.
- Cook, E. R., Meko, D. M., Stahle, D. W., & Cleaveland, M. K. (1999). Drought reconstructions for the continental United States. *Journal of Climate*, 12, 1145–1162.
- Fritts, H. C. (1976). Tree rings and climate. London: Academic Press.
- Giralt, S., Julia, R., Klerkx, J., Riera, S., Leroy, S., Buchaca, T., et al. (2004). 1000-Year environmental history of Lake Issyk-Kul. In J. C. J. Nihoul, P. O. Zavialov, & P. P. Micklin (Eds.), *Dying and dead seas: Climatic versus anthropic causes* (pp. 253–285). Dordrecht: Kluwer Academic Publishers.
- Gou, X. H., Deng, Y., Chen, F. H., Yang, M. X., Fang, K. Y., Gao, L., et al. (2010). Tree-ring based streamflow reconstruction for the Upper Yellow River over the past 1234 years. *Chinese Science Bulletin*, 55, 4179–4186.
- Grove, J. M. (2001). The initiation of the "Little Ice Age" in regions round the North Atlantic. Climatic Change, 48(1), 53-82.
- Hedrick, K. A., Seong, Y. B., Owen, L. A., Caffee, M. W., & Dietsch, C. (2011). Toward defining the transition in style and timing of Quaternary glaciation between the monsoon-influenced Greater Himalaya and the semi-arid transhimalaya of Northern India. *Quaternary International*, 236(1–2), 21–33.
- Holmes, R. L. (1983). Computer-assisted quality control in tree-ring dating and measurement. Tree-Ring Bulletin, 43, 69-78.

Hughes, M. K. (1992). Dendroclimatic evidence from western Himalaya. In R. S. Bradley, & P. D. Jones (Eds.), *Climate since CE 1500* (pp. 415–431). London: Routledge.

Hughes, M. K. (2001). An improved reconstruction of summer temperature at Srinagar, Kashmir since 1660 CE, based on tree ring width and maximum latewood density of Abies pindrow (Royle) Spach. *Palaeobotanist*, 50, 13–19.

Hughes, M. K., & Diaz, H. F. (1994). Was there a "Medieval Warm Period," and if so, where and when? Climatic Change, 26(2–3), 109–142. Jones, P. D., Briffa, K. R., Barnett, T. P., & Tett, S. F. B. (1998). High-resolution Paleoclimatic records for the last Millennium: Interpretation, integration and comparison with general circulation model control run temperatures. *The Holocene*, 8, 455–471.

- Kotlia, B. S., Singh, A. K., Zhao, J. X., Duan, W., Tan, M., Sharma, A. K., & Raza, W. (2017). Stalagmite based high resolution precipitation variability for past four centuries in the Indian Central Himalaya: Chulerasim cave re-visited and data re-interpretation. *Quaternary International*, 444, 35–43.
- Koul, A. (1978). Geography of the Jammu and Kashmir State (revised by P. N. K. Bamzai). New Delhi: Light and Life Publishers.
- Kulkarni, A. V., & Alex, S. (2003). Estimation of recent glacial variations in Baspa Basin using remote sensing technique. Journal of the Indian Society of Remote Sensing, 31(2), 81–90.
- Kulkarni, A. V., & Rathore, B. P. (2003). Snow cover monitoring in Baspa basin using IRS WiFS data. Mausam, 54, 335–340.
- Kulkarni, A. V., Rathore, B. P., & Singh, S. K. (2010). Distribution of seasonal snow cover in central and western Himalaya. Annals of Glaciology, 51(54), 123–128.
- Lehmkuhl, F., Owen, L. A., & Derbyshire, E. (1998). Late Quaternary glacial history of northeast Tibet. *Journal of Quaternary Science*, 13(6), 121–142.
- Managave, S., Shimla, P., Yadav, R. R., Ramesh, R., & Balakrishnan, S. (2020). Contrasting centennial-scale climate variability in high mountain Asia revealed by a tree-ring oxygen isotope record from Lahaul-Spiti. *Geophysical Research Letters*, 47(4), e2019GL086170. https://doi.org/10.1029/2019GL086170
- Mann, M. E. (2002). Little Ice Age. Encyclopedia of global environmental change, 1, 504-509.
- Mann, M. E., Bradley, R. S., & Hughes, M. K. (1998). Global-scale temperature patterns and climate forcing over the past six centuries. *Nature*, 392, 779–787.
- Mann, M. E., Bradley, R. S., & Hughes, M. K. (1999). Northern Hemisphere temperatures during the past millennium: Inferences, uncertainties, and limitations. *Geophysical Research Letters*, 26, 759–762. https://doi.org/10.1029/1999GL900070
- Mann, M. E., Zhang, Z., Rutherford, S., Bradley, R. S., Hughes, M. K., Shindell, D., et al. (2009). Global signatures and dynamical origins of the Little Ice Age and Medieval Climate Anomaly. Science, 326, 1256–1260.
- Matthews, J. A., & Briffa, K. R. (2005). The "Little Ice Age": Re-evaluation of an evolving concept. Geografiska Annaler Series A: Physical Geography, 87(1), 17–36.
- Meko, D. (1997). Dendroclimatic reconstruction with time varying predictor subsets of tree indices. Journal of Climate, 10, 687-696.
- Michaelsen, J. (1987). Cross-validation in statistical climate forecast models. Journal of Climate and Applied Meteorology, 26, 1589–1600.
 Miller, G. H., Geirsdóttir, Á., Zhong, Y., Larsen, D. J., Otto-Bliesner, B. L., Holland, M. M., et al. (2012). Abrupt onset of the Little Ice Age triggered by volcanism and sustained by sea-ice/ocean feedbacks. Geophysical Research Letters, 39, L02708. https://doi.
- org/10.1029/2011GL050168 Misra, K. G., Yadav, R. R., & Misra, S. (2015). Satluj river flow variations since CE 1660 based on tree-ring network of Himalayan cedar from
- western Himalaya, India. *Quaternary International*, 371, 135–143. Murari, M. K., Owen, L. A., Dortch, J. M., Caffee, M. W., Dietsch, C., Fuchs, M, et al. (2014). Timing and climatic drivers for glaciation
- across monsoon-influenced regions of the Himalayan-Tibetan orogen. Quaternary Science Reviews, 88, 159–182.
 Nesje, A., & Dahl, S. O. (2003). The "Little Ice Age"-only temperature? The Holocene, 13(1), 139–145.

Owen, L. A., Finkel, R. C., Barnard, P. L., Haizhou, M., Asahi, K., Caffee, M. W., & Derbyshire, E. (2005). Climatic and topographic controls on the style and timing of Late Quaternary glaciation throughout Tibet and the Himalaya defined by 10Be cosmogenic radionuclide surface exposure dating. *Quaternary Science Reviews*, 24(12–13), 1391–1411.

Owen, L. A., Gualtieri, L. Y. N., Finkel, R. C., Caffee, M. W., Benn, D. I., & Sharma, M. C. (2001). Cosmogenic radionuclide dating of glacial landforms in the Lahul Himalaya, northern India: Defining the timing of Late Quaternary glaciation. *Journal of Quaternary Science: Published for the Quaternary Research Association*, 16(6), 555–563.

Owen, L. A., Kamp, U., Spencer, J. Q., & Haserodt, K. (2002). Timing and style of Late Quaternary glaciation in the eastern Hindu Kush, Chitral, northern Pakistan: A review and revision of the glacial chronology based on new optically stimulated luminescence dating. *Quaternary International*, 97, 41–55.

Owen, L. A., Scott, C. H., & Derbyshire, E. (2000). The quaternary glacial history of Nanga Parbat. Quaternary International, 65, 63–79.

PAGES 2k Consortium. (2013). Continental-scale temperature variability during the past two millennia. *Nature Geoscience*, 6(5), 339–346. Porter, S. C. (1981). Lichenometric studies in the Cascade Range of Washington: Establishment of Rhizocarpon geographicum growth curves at Mount Rainier. *Arctic and Alpine Research*, *13*(1), 11–23.

Qiao, B., & Yi, C. (2017). Reconstruction of Little Ice Age glacier area and equilibrium line attitudes in the central and western Himalaya. *Quaternary International*, 444, 65–75.

Rao, M. P., Cook, E. R., Cook, B. I., Palmer, J. G., Uriarte, M., Devineni, N., et al. (2018). Six centuries of Upper Indus Basin streamflow variability and its climatic drivers. *Water Resources Research*, 54, 5687–5701. https://doi.org/10.1029/2018WR023080

Rinn, F. (2003). TSAP-win time series analysis and presentation for dendrochronology and related applications, Version 0.53 for Microsoft Windows. Heidelberg: Rinn Tech.

Rowan, A. V. (2017). The "Little Ice Age" in the Himalaya: A review of glacier advance driven by Northern Hemisphere temperature change. *The Holocene*, 27, 292–308.

Saini, R., Sharma, M. C., Deswal, S., Barr, I. D., Kumar, P., Kumar, P., et al. (2019). Glacio-archeological evidence of permanent settlements within a glacier end moraine complex during 980–1840 CE: The Miyar Basin, Lahaul Himalaya, India. *Anthropocene*, *26*, 100197.

Shah, S. K., Pandey, U., & Mehrotra, N. (2018). Precipitation reconstruction for the Lidder Valley, Kashmir Himalaya using tree-rings of Cedrus deodara. International Journal of Climatology, 38, e758–e773.

Shao, X., Huang, L., Liu, H., Liang, E. Y., Fang, X. Q., & Wang, L. L. (2005). Reconstruction of precipitation variation from tree rings in recent 1000 years in Delingha, Qinghai. Science in China – Series D: Earth Sciences, 48, 939–949.

Shekhar, M., Bhardwaj, A., Singh, S., Ranhotra, P. S., Bhattacharyya, A., Pal, A. K., et al. (2017). Himalayan glaciers experienced significant mass loss during later phases of Little Ice Age. Scientific Reports, 7(1), 1–14.

Singh, J., Park, W. K., & Yadav, R. R. (2006). Tree-ring-based hydrological records for western Himalaya, India, since CE 1560. Climate Dynamics, 26, 295–303.

Singh, J., & Yadav, R. R. (2013). Tree-ring-based seven century long flow records of Satluj River, western Himalaya, India. Quaternary International, 304, 156–162.

Singh, V., Yadav, R. R., Gupta, A. K., Kotlia, B. S., Singh, J., Yadava, A. K., & Misra, K. G. (2017). Tree ring drought records from Kishtwar, Jammu and Kashmir, northwest Himalaya, India. *Quaternary International*, 444, 53–64.

Solomina, O. N., Bradley, R. S., Hodgson, D. A., Ivy-Ochs, S., Jomelli, V., Mackintosh, A. N, et al. (2015). Holocene glacier fluctuations. *Quaternary Science Reviews*, 111, 9–34.

Stokes, M. A., & Smiley, T. L. (1968). An introduction to tree-ring dating. Chicago, IL: The University of Chicago Press.

Thompson, L. G., Mosley-Thompson, E., Davis, M. E., Lin, P. N., Dai, J., Bolzan, J. F., & Yao, T. (1995). A 1000-year climate ice-core record from the Guliya ice cap, China: Its relationship to global climate variability. *Annals of Glaciology*, 21, 175–181.

Thompson, L. G., Yao, T., Davis, M. E., Henderson, K. A., Mosley-Thompson, E., Lin, P. N., et al. (1997). Tropical climate instability: The last glacial cycle from a Qinghai-Tibetan ice core. *Science*, *276*, 1821–1825.

Treydte, K. S., Schleser, G. H., Helle, G., Frank, D. C., Winiger, M., Haug, G. H., & Esper, J. (2006). The twentieth century was the wettest period in northern Pakistan over the past millennium. *Nature*, 440, 1179–1182.

Yadav, R. R. (2011). Long-term hydroclimatic variability in monsoon shadow zone of western Himalaya, India. Climate Dynamics, 36, 1453–1462.

Yadav, R. R. (2011). Tree ring evidence of a 20th century precipitation surge in the monsoon shadow zone of the western Himalaya, India. Journal of Geophysical Research, 116(D2). https://doi.org/10.1029/2010JD014647

Yadav, R. R., & Bhutiyani, M. R. (2013). Tree-ring-based snowfall record for cold arid western Himalaya, India since CE 1460. Journal of Geophysical Research: Atmosphere, 118, 7516–7522. https://doi.org/10.1002/jgrd.50583

Yadav, R. R., Braeuning, A., & Singh, J. (2011). Tree-ring-inferred summer temperature variations over the last millennium in western Himalaya, India. Climate Dynamics, 36, 1545–1554.

Yadav, R. R., Gupta, A. K., Kotlia, B. S., Singh, V., Misra, K. G., Yadava, A. K., & Singh, A. K. (2017). Recent wetting and glacier expansion in the Northwest Himalaya and Karakoram. *Scientific Reports*, 7, 6139.

Yadav, R. R., Park, W.-K., Singh, J., & Dubey, B. (2004). Do the western Himalayas defy global warming? *Geophysical Research Letters*, *31*, L17201. https://doi.org/10.1029/2004GL020201

Yadava, A. K., Braeuning, A., Singh, J., & Yadav, R. R. (2016). Boreal spring precipitation variability in the cold arid western Himalaya during the last millennium, regional linkages, and socio-economic implications. *Quaternary Science Reviews*, 144, 28–43.

Zhang, Q. B., Cheng, G., Yao, T., Kang, X., & Huang, J. (2003). A 2,326-year tree-ring record of climate variability on the northeastern Qinghai-Tibetan Plateau. *Geophysical Research Letters*, 30, 1739–1742. https://doi.org/10.1029/2003GL017425

Zheng, J., Xiao, L., Fang, X., Hao, Z., Ge, Q., & Li, B. (2014). How climate change impacted the collapse of the Ming dynasty. *Climatic Change*, 127, 169–182.

References From the Supporting Information

Cook, E. R. (1985). A time series analysis approach to tree-ring standardization (Ph.D. thesis). Tucson, AZ: University of Arizona.

Cook, E. R., & Peters, K. (1981). The smoothing spline: A new approach to standardizing forest interior tree-ring width series for dendroclimatic studies. *Tree-Ring Bulletin*, 41, 45–53.

Cook, E. R., & Peters, K. (1997). Calculating unbiased tree-ring indices for the study of climatic and environmental change. *The Holocene*, 7, 361–370.



Melvin, T. M., & Briffa, K. R. (2008). A "Signal-Free" approach to dendroclimatic standardisation. *Dendrochronologia*, 26, 71–86.
 Wigley, T. M. L., Briffa, K. R., & Jones, P. D. (1984). On the average value of correlated time series, with applications in dendroclimatology and hydrometeorology. *Journal of Climate and Applied Meteorology*, 23, 201–213.